Author's response to reviews

Title: Are modern contraceptives acceptable to people and where do they source them from across Nigeria?

Authors:

Jane C Enemuoh (chinelo.enemuoh@yahoo.com)
Obinna E Onwujekwe (onwujekwe@yahoo.co.uk)
Chinyere Mbachu (chinyere23ng@yahoo.com)
Chinwe Ogbonna (ogbonna@unfpa.org)
Benjamin SC Uzochukwu (bscuzochukwu@yahoo.com)
Agathe Lawson (alawson@unfpa.org)
Bannet Ndyanabangi (ndyabangi@unfpa.org)

Version: 3 Date: 19 May 2012

Author's response to reviews:

The revision made in the manuscript was inclusion of ethical committee that approved the research and the type of consent obtained from respondent. The ethical approval was obtained from the University of Nigeria Teaching Hospital Enugu (UNTH) written consent was obtained from the respondent before interview comensed.